Author’s response to reviews

Title: Perceptions of Isolation during Facility Births in Haiti - A Qualitative Study

Authors:
Alka Dev (alka.dev@dartmouth.edu)
Chelsey Kivland (chelsey.l.kivland@dartmouth.edu)
Mikerlyne Faustin (mikerlynefaustin@yahoo.fr)
Olivia Turnier (olivia_turnier@hotmail.com)
Tatiana Bell (tbell@gheskio.org)
Marie Denise Leger (denisesamsonht@yahoo.fr)

Version: 2 Date: 08 Oct 2019

Author’s response to reviews:

October 8, 2019

Re: REPH-D-19-00245R1
Perceptions of Isolation during Facility Births in Haiti - A Qualitative Study
Alka Dev, DrPH; Chelsey Kivland, PhD; Mikerlyne Faustin; Olivia Turnier; Tatiana Bell; Marie Denise Leger, MA

Responses to Reviewer Comments regarding copy editing:

The entire paper needs to be copy-edited by a professional science writer, or one of the authors with those skills.

Response: We have had the paper copy edited by a professional copy-editing service. Tracked changes are attached.

I am using just two examples from the first two pages, but errors are found through out. Examples are lack of agreement between pronouns and nouns/verbs and awkward constructions, but there are more, which must be addressed.

This has been addressed throughout the paper.

Please make a clean copy accepting all of the changes in submission 1, but tracking all of your copy edits.

Completed.

Sincerely,
Alka Dev